Author’s response to reviews

Title: The relationship between serum fatty-acid binding protein 4 level and lung function in Korean subjects with normal ventilatory function

Authors:

Hye-Jeong Park (hj207.park@samsung.com)
Se Eun Park (liplips@hanmail.net)
Cheol-Young Park (cyber.park@samsung.com)
Won-Young Lee (drlwy@hanmail.net)
Seong Yong Lim (mdlimsy@skku.edu)
Ki-Won Oh (okwendo@naver.com)
Sung-Woo Park (sungwoo0913.park@samsung.com)
Eun-Jung Rhee (hongsiri@hanmail.net)

Version: 2 Date: 04 Jan 2016

Author’s response to reviews:

1) Ethics: the 1975 Declaration of Helsinki has been updated in recent years. Please clarify whether your study meets the requirements of the latest update of the Declaration.

: we've corrected the 1975 -> to 2000, the first revision year of Helsingki Declaration

2) Textural duplication flagged by iThenticate: there are a number of portions with your paper, particularly within the Background, Methods and Discussion sections, that have been taken directly from previous publication. Please could you look through your manuscript and rephrase these sections accordingly using your own wording.
Thank you for your kind comment. We’ve tried to change the manuscript into our own wording in introduction and discussion, but in the method section, the method could be very similar to the previously published paper of our own. We hope that you could understand.